Author's response to reviews

**Title:** Circulating miRNAs reflect early myocardial injury and recovery after heart transplantation

**Authors:**

Enshi Wang (enshi.wang@163.com)
Yu Nie (niyuniverse@126.com)
Zhe Zheng (zhengzhe@fuwai.com)
Jie Huang (jiehuangpumc@sina.cn)
Hao Zhang (drzhanghao@yahoo.com)
Shengshou Hu (shengshouh@sina.cn)

**Version:** 2  **Date:** 21 February 2013

**Author's response to reviews:** see over
Dear Editor:
There are some problems on author name, author contribution, and fund support. So we made some revision as below.

1. We add Wei Wang and Zhongkai Liao in our author list because of their contribution to the work. Then Zhe Zheng were added to the co-corresponding author.
2. This work was supported by Chinese government “863” program “Key technology of heart transplantation and immune regulation”. No:2012AA021009. So we added this part into acknowledgement.

Sincerely yours,
Shengshou Hu

ADDRESS: Department of Cardiac Surgery, State Key Laboratory of Cardiovascular Disease, Fuwai Hospital, Chinese Academy of Medical Sciences, Xicheng District, Beijing, 100037, China

Email: shengshouh@sina.cn
Tel & Fax: 86-10-88396050